Author’s response to reviews

Title: “Strong Teeth” – a study protocol for an early-phase feasibility trial of a complex oral health intervention delivered by dental teams to parents of young children

Authors:

K Tull (k.tull@leeds.ac.uk)
Kara Gray-Burrows (K.Gray-Burrows@leeds.ac.uk)
Amrit Bhatti (a.bhatti@leeds.ac.uk)
Jenny Owen (j.c.owen@leeds.ac.uk)
Lucy Rutter (L.M.Rutter@leeds.ac.uk)
Timothy Zoltie (T.Zoltie@leeds.ac.uk)
Jayne Purdy (J.Purdy@leeds.ac.uk)
Erin Giles (E.Giles@leeds.ac.uk)
Carron Paige (C.J.Paige@leeds.ac.uk)
Morvin Patel (morvin.patel1@nhs.net)
Zoe Marshman (z.marshman@sheffield.ac.uk)
Robert West (r.west@leeds.ac.uk)
Sue Pavitt (s.pavitt@leeds.ac.uk)
Peter Day (p.f.day@leeds.ac.uk)

Version: 3 Date: 24 Jul 2019

Author’s response to reviews:

We would like to thank the editor for his recommendation to register our feasibility protocol. The study is now registered in the ISRCTN Register (ISRCTN10709150), and this information has been added to the abstract of the manuscript.